Author's response to reviews

Title: Early uptake of HIV counseling and testing among pregnant women at different levels of health facilities - experiences from a community-based study in Northern Vietnam

Authors:

Hanh Nguyen (n_thuyhanh@yahoo.com)
Tine Gammeltoft (tine.gammeltoft@anthro.ku.dk)
Vibeke Rasch (vira@sund.ku.dk)

Version: 9 Date: 24 December 2010

Author's response to reviews: see over
Letter to the Editor (15th October 2010)

Authors of
Manuscript ID: 1277418933240516
Title: Uptake of HIV counselling and testing among pregnant women at different levels of health facilities - experiences from a community-based study in Northern Vietnam

Journal: BMC Health Services Research
Submitted: 11 Nov 2009

First of all, we would like to thank the Editor and the Reviewer for commenting on the 4th version of our manuscript. We have revised the manuscript accordingly. Please note that the page numbers given refer to page numbers in the revised and resubmitted version of the manuscript. We have used track changes to show all the changes of the manuscript.

Comment 1:
Response:

Comment 2:
Response:

Thanks for your attention and looking forward to hearing from you
Best regards,

Vibeke Rasch, Tine Gammeltoft and Nguyen Thi Thuy Hanh